A duodenal sleeve bypass device added to intensive medical therapy for obesity with type 2 diabetes: a RCT

Aruchuna Ruban,^{1*} Michael A Glaysher,²
Alexander D Miras,³ Anthony P Goldstone,⁴
Christina G Prechtl,⁵ Nicholas Johnson,⁵ Jia Li,¹
Madhawi Aldhwayan,³ Ghadah Aldubaikhi,⁴
Ben Glover,¹ Joanne Lord,⁶ Olu Onyimadu,⁶
Emmanuela Falaschetti,⁵ Natalia Klimowska-Nassar,⁵
Hutan Ashrafian,¹ James Byrne⁷ and Julian P Teare¹

Declared competing interests of authors: Christina G Prechtl received subvention funding approval from the Department of Health and Social Care and funding from Nutricia (Zoetermeer, the Netherlands) for nutritional drinks. Aruchuna Ruban received travel fees from GI Dynamics Inc. (Boston, MA, USA). Alexander D Miras has received honoraria for presentations and advisory board contributions by Novo Nordisk (Bagsværd, Denmark), Boehringer Ingelheim (Ingelheim am Rhein, Germany), AstraZeneca (Cambridge, UK), Ethicon (Johnson & Johnson, Brunswick, NJ, USA) and research grant funding from Fractyl (Lexington, MA, USA). Anthony P Goldstone is on a data safety monitoring committee for clinical trials in obesity for Novo Nordisk. Joanne Lord is a member of the Health Technology Assessment Stakeholder Advisory Group (2015–20). Julian P Teare received travel fees from GI Dynamics Inc.

Published November 2020

DOI: 10.3310/eme07060

¹Department of Surgery and Cancer, St Mary's Hospital, Imperial College London, London, UK

²National Institute for Health Research Southampton Biomedical Research Centre, Southampton Centre for Biomedical Research, Southampton General Hospital, Southampton, UK

³Section of Investigative Medicine, Division of Diabetes, Endocrinology and Metabolic Medicine, Hammersmith Hospital, Imperial College London, London, UK

⁴PsychoNeuroEndocrinology Research Group, Neuropsychopharmacology Unit, Centre for Psychiatry and Computational, Cognitive and Clinical Neuroimaging Laboratory, Division of Brain Sciences, Imperial College London, London, UK

⁵Department of Public Health, Imperial Clinical Trials Unit, Imperial College London, London, UK

⁶University of Southampton, Southampton Health Technology Assessments Centre, University of Southampton Science Park, Southampton, UK

⁷Division of Surgery, University Hospital Southampton NHS Foundation Trust, Southampton, UK

^{*}Corresponding author a.mohanaruban@imperial.ac.uk

Plain English summary

A duodenal sleeve bypass device for obesity with type 2 diabetes Efficacy and Mechanism Evaluation 2020; Vol. 7: No. 6

DOI: 10.3310/eme07060

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

besity is a worldwide issue and is associated with complications such as type 2 diabetes mellitus. Both conditions cause suffering to the individual and are costly for health-care services. The duodenal-jejunal sleeve bypass (EndoBarrier®; GI Dynamics Inc., Boston, MA, USA) is a removable tubular device that is implanted using a telescope into the small intestine via the mouth and stomach without the need for surgery.

In this study, 170 patients were recruited across two hospital sites. All participants received lifestyle modification advice and optimisation of their diabetes control with intensive medical therapy. Half of the patients also received the endoluminal duodenal–jejunal bypass liner for 12 months and then all patients were seen periodically over a further 12-month period to have their progress monitored. The main aim of this study was to identify the proportion of patients who achieved glucose improvement as defined by the International Diabetes Federation. The study also investigated the degree of weight loss achieved and the safety of the device. To understand how the endoluminal duodenal–jejunal bypass liner works, additional studies were conducted to measure the brain response to food using brain imaging techniques, body sensitivity to insulin and food choices, and to analyse the breakdown products produced by bacteria in our bodies.

Both treatments produced similar improvements in glucose control, but people in the endoluminal duodenal-jejunal bypass liner group lost more weight and achieved better blood pressure control than those in the control group while the device was in place, but not after its removal. Significant reactions and side effects occurred more frequently in the endoluminal duodenal-jejunal bypass liner group than in the group that did not receive the device, and the majority of these side effects were deemed to be definitely related to the duodenal-jejunal bypass liner. At 6 months, the degree to which the body's cells respond to the hormone insulin and take up glucose from the blood was improved in the duodenal-jejunal bypass liner group. There were no differences in brain responses to food or eating behaviour between the groups. Breakdown products of metabolism detected in blood, urine and faeces were found to vary significantly between the treatment groups.

Overall, these results indicate that the endoluminal duodenal–jejunal bypass liner is not better than intensive medical therapy for glucose control. We showed some benefit to weight loss at 1 year but not at 2 years. The evidence suggests that the device does not appear to be a cost-effective strategy for glucose control or weight loss.

Efficacy and Mechanism Evaluation

ISSN 2050-4365 (Print)

ISSN 2050-4373 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full EME archive is freely available to view online at www.journalslibrary.nihr.ac.uk/eme. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Efficacy and Mechanism Evaluation journal

Reports are published in *Efficacy and Mechanism Evaluation* (EME) if (1) they have resulted from work for the EME programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

EME programme

The Efficacy and Mechanism Evaluation (EME) programme funds ambitious studies evaluating interventions that have the potential to make a step-change in the promotion of health, treatment of disease and improvement of rehabilitation or long-term care. Within these studies, EME supports research to improve the understanding of the mechanisms of both diseases and treatments.

The programme supports translational research into a wide range of new or repurposed interventions. These may include diagnostic or prognostic tests and decision-making tools, therapeutics or psychological treatments, medical devices, and public health initiatives delivered in the NHS.

The EME programme supports clinical trials and studies with other robust designs, which test the efficacy of interventions, and which may use clinical or well-validated surrogate outcomes. It only supports studies in man and where there is adequate proof of concept. The programme encourages hypothesis-driven mechanistic studies, integrated within the efficacy study, that explore the mechanisms of action of the intervention or the disease, the cause of differing responses, or improve the understanding of adverse effects. It funds similar mechanistic studies linked to studies funded by any NIHR programme.

The EME programme is funded by the Medical Research Council (MRC) and the National Institute for Health Research (NIHR), with contributions from the Chief Scientist Office (CSO) in Scotland and National Institute for Social Care and Health Research (NISCHR) in Wales and the Health and Social Care Research and Development (HSC R&D), Public Health Agency in Northern Ireland.

This report

The research reported in this issue of the journal was funded by the EME programme as project number 12/10/04. The contractual start date was in May 2014. The final report began editorial review in June 2019 and was accepted for publication in March 2020. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The EME editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research. The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the MRC, NETSCC, the EME programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the EME programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2020. This work was produced by Ruban *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

Editor-in-Chief of **Efficacy and Mechanism Evaluation** and NIHR Journals Library

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGFAR, PHR journals) and Editor-in-Chief of HS&DR, PGFAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk